Author's response to reviews

Title: Oxaliplatin-based versus irinotecan-based hyperthermic intraperitoneal chemotherapy (HIPEC) in patients with colorectal peritoneal metastasis: a retrospective analysis

Authors:

Gabriel Glockzin (gabriel.glockzin@ukr.de)
Michael Gerken (michael.gerken@ukr.de)
Sven A Lang (sven.lang@ukr.de)
Monika Klinkhammer-Schalke (monika.klinkhammer-schalke@ukr.de)
Pompiliu Piso (pompiliu.piso@barmherzige-regensburg.de)
Hans J Schlitt (hans.schlitt@ukr.de)

Version: 2
Date: 24 April 2014

Author's response to reviews:

1) Line numbering has been added to the revised manuscript.
2) 'Competing interests' and 'Abbreviations' have been added to the revised manuscript.
3) Figures 1A and 1B have been summarized in one page.
4) Acknowledgements have been indicated as 'none'.